Author's response to reviews

Title: Women's Evaluation of Abuse and Violence Care in General Practice: a Cluster Randomised Controlled Trial (weave)

Authors:

Kelsey L Hegarty (k.hegarty@unimelb.edu.au)
Jane M Gunn (j.gunn@unimelb.edu.au)
Lorna J O'Doherty (l.odoherty@uimelb.edu.au)
Angela Taft (a.taft@latrobe.edu.au)
Patty Chondros (p.chondros@uimelb.edu.au)
Gene Feder (gene.feder@bristol.ac.uk)
Jill Astbury (jill.astbury@vu.edu.au)
Stephanie Brown (stephanie.brown@mcri.edu.au)

Version: 3 Date: 23 December 2009

Author's response to reviews:

Dear Editor

Thank you for allowing us to resubmit. We have as requested altered the

1. Author list: to “First name Initial Last name”

2. Structure: we have reformatted it to use these headings: Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors’ contributions; Authors’ information (optional); Acknowledgements (optional); References

3. moved [ACTRN12608000032358 to the end of the abstract.

4. Figure and table placement text: removed this.

5. Tables: moved title to above the table

6. Fixed some minor typos.

Thanks

Kelsey Hegarty